---
title: "R at Kellogg"
---

```{r, include=FALSE}
sectiontabs <- '.tabset .tabset-fade .tabset-pills'
##sectiontabs <- ''
```


The purpose of this website ([source](index.Rmd)) is to share
material that may be useful across courses. zzz

## Getting started {`r sectiontabs`}

### New to R?

* Install instructions for R, RStudio, and LaTeX ([html](R_RStudio_Software_Setup.html))
  ([pdf](R_RStudio_Software_Setup.pdf)) ([Rmd](R_RStudio_Software_Setup.Rmd))
  
### Already have R?

* It's great that you already have R -- you also need RStudio. They are
  not the same thing: R is the programming language and RStudio is a
  visual interface that makes it easier to work in R. [You can download
  RStudio by clicking
  here.](https://www.rstudio.com/products/rstudio/download/)

* Make sure R is up-to-date and that you have installed additional
  required packages. It's also helpful to install
  LaTeX. ([html](R_RStudio_Update.html)) ([pdf](R_RStudio_Update.pdf))


## Learning R

* Professor Gordon's tutorial for Retail Analytics (introductory):
  ([html](R_Kellogg_Tutorial.html)) ([pdf](R_Kellogg_Tutorial.pdf))
  

* Understanding dataframe operations (intermediate) ([html](dataframe_ops.html))
  ([pdf](dataframe_ops.pdf))  ([Github](https://github.com/rmcd1024/Rinstruction))


##  Videos {`r sectiontabs`}

### RStudio orientation

Scripts for the videos below ([pdf](pre-class-videos.pdf), [html](pre-class-videos.html))

1. [Introduction to RStudio](video/Rstudio\ intro.mov)
2. [Introduction to the Rstudio Editor](video/Rstudio\ Editor\ Intro.mov)
3. [Introduction to Vectors](video/vectors.mov)
 

### Introduction to R

1. [Intro](https://www.screencast.com/t/q4A8pGgdNFZ)
1. [Basics](https://www.screencast.com/t/qXUQzCcoy)
1. [Manipulating Data Frames](https://www.screencast.com/t/SQbTUExtHs)
1. [Plots](https://www.screencast.com/t/2fPtFEY1BjJH)
1. [Regression](https://www.screencast.com/t/qFAXj36RzQXk)

## Course-specific materials

* [DECS-922: Data Exploration](mcdonald)

* [MKTG-462: Retail Analytics](gordon)

* [MKTG-482: Customer Analytics](zettelmeyer)


<!--
* [MTKT-955: Digital Marketing Analytics](cutler)
-->

* [STRT-469: Analytics for Strategy](prager)

<!--

* [OPNS-451: Analytical Decision Modeling: R](opns451.html)

-->
